Author's response to reviews

Title: Do pregnant women's socioeconomic status influence the relationship between use of selective serotonin-reuptake inhibitors and the risk of congenital heart defects? - a cohort study

Authors:

Tanja M Knudsen (tan.knudsen@hotmail.com)
Anne V Hansen (avih@sund.ku.dk)
Ester Garne (egarne@dadlnet.dk)
Anne-Marie N Andersen (amny@sund.ku.dk)

Version: 2
Date: 19 March 2014

Author's response to reviews:

Dear Editor

According to your request, I have made the following changes to the manuscript.
- Changed the section ethics, so it meets your requirements.
- Included line numbers in the main text of the manuscript.

Yours sincerely
Tanja Majbrit Knudsen